Author’s response to reviews

Title: Parents and nurses telling their stories: The perceived needs of parents caring for critically ill children at Kilimanjaro Christian Medical Centre in Tanzania

Authors:

Vivian Frank Saria (vivianfrank@gmail.com)
Lilian Teddy Mselle (nakutz@yahoo.com)
Birgit Anne Siceloff (bjorgensen59@gmail.com)

Version: 4 Date: 28 Oct 2019

Author’s response to reviews:

NURS-D-19-00004R3
Parents and nurses telling their stories: Perceived needs of parents caring for critically ill children at Kilimanjaro Christian Medical Centre in Tanzania

Vivian Frank Saria, MScN, BScN, RN; Lilian Teddy Mselle, PhD, Mphil, BA, RNM; Birgit Anne Siceloff, MPH, BScN, RN

1. I'm afraid your manuscript will need additional language editing. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.
Response: The manuscript has been English edited by the native English speakers as suggested

2. Please be sure to include a consent statement in the Ethics Approval and Consent to Participate section.
Response: The consent statement is included in the Ethics Approval and Consent to Participate section as requested. Page 22 line 22 to page 23 line 1

3. In the main text of your manuscript, we note that you have indicated that written consent was obtained and later stated that verbal consent was obtained. Please clarify the form of consent and specific the cases when written or verbal consent was obtained, if appropriate.
Response: Written consent was obtained from the participants to be interviewed and verbal consent was given by participant for their voices to be recorded. This information is clearly written in the manuscript. Page 6 line 23 and Page 8 line 4

4. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).
Response: The description of the contribution done by BAS in manuscript preparation has been revised based on authorship guideline. Page 23 line 20-21.

5. Please upload English language versions of all questionnaires and focus group guides developed specifically for use in this study as supplementary files. If the questionnaires and/or focus group guides are not yours or are already published by you, please supply references and/or links to them.
Response: The English language versions of the focus group guide used has been uploaded as additional materials (Additional file 1 &amp; 2) as suggested.

6. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.
Response: We have revised the manuscript and the responses is uploaded in the designated input box as suggested.